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PSB

Main changes during EYETS 2016/17

* New turn-by-turn ring PU system (BTMS):

e Commissioning during 2017 startup

* Working well now; some issue with reboots (requires
specialist) and electronics overheating still to be solved

 Calibration issue? (tbc by Jeroen)

* Prototype new wire scanner in R3
 Commissioning started 21/06

* New Flat lonisation Chambers (FICs)
 Commissioning still to be done

* New wide-band pickup in PSB-to-PS transfer line

* Commissioning still to be done B. Mikulec
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PSB

PSB Wire Scanners
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PSB

Other Operational Issues

e BTP.BPM20: ISOLDE beam to dump (no beam in BTP...)
* Oscillations mainly in hor. S
plane related to beam passage . — _‘ .
(even beam not being sent to

PS); present since at least
March 2016; intervention on
amplifier and cables planned
during ITS2

* Tune measurement

* Kick is shifting for rings 1+2;
workaround in place, but 2 ; ~ . 2
should be understood L1l 2 00 8w e

B. Mikulec
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PSB

Summary PSB

* Main worry with wire scanners

* BTMS, FICs, prototype new wire scanner and wide-band PU
commissioning/optimisation to be finished

* Good support from Bl — thanks a lot!

* Awaiting installation of new tune PUs and R3 matching
monitor during YETS 2017/18

* HST extremely useful for instrument evaluations

* Check of full functionality of H%/H- monitor only outstanding point
* Non-related BCT issue could be understood and solved

* Make sure through good continued collaboration Bl — OP to
be in a comfortable situation during restart after LS2
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PS

Wire Scanners

* intense use during 2016/17 for systematic measurement of the
LHC beams and to verify the islands of the MTE beam.

* future development of bunch-by-bunch measurement (2018-
LS27?)

e OASIS signals for new generation under discussion

(& -= Easy Wirescanner =- - CPS.USER.MD3
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PS

PUINnTT2:

 survey of spill quality of the MTE beam (5 turns-islands-core)
using PU 208/228 during 2016/17

e waiting for a signal (sum and frequency) of a PU downstream of
the correctors DFA(242, 254) during 2017 run

= Inspector 2.2.38"- HIMTE_Vista

HIiMTE Spill Monitor  ERaGEluly

D. Cotte
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PS

BGI82Hz :
 first promising results in 2017; waiting for MD slots in order to

better understand the system; in particular cancel the integrated
magnetic field seen by the beam
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PS

BLMs:

» good performance in 2016/17; all ACEMs working at PS start-up

* waiting for availability of the 17 diamond detectors to check
shadowing between TPS15 and SMH16 for MTE beam (end 2017)

* first tests with 100 LHC-type BLMs in progress 2017

Ring PU:

» good performance in 2016/17

* study possibility to have a dedicated tune PU cabled in
qguadrupolar mode without impacting tune measurement of
small beams

D. Cotte
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PSB/PS

BWS measurements figure of merit needed:

* Noteasy tojudge if a scan is well done in terms of PM saturation and optical filter choice
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Clear messages in the application:
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* Wireresistance??
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EAST Area

improvement of the SPILL F61 signal in order to make better use of the ADC

resently around 1%
(p y )

add BLMs/PU in F61, F61S, F61N to facilitate the steering and avoid hot spots
(e.g. the one measured at F61.BHZ01 during TS1)
the BPMs in T8 are very useful to do the steering!
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ISOLDE

ISOLDE BEAM DIAGNOSTICS:

ISOLDE Beam Diagnostics
High Level Control Software

Separators: N

= 8 x Faraday Cups £ el ewlor T w‘“‘”‘” T Commmme o] mesonmn ovsrcos (56 ot

= 7 x Semgrids R e Ll gy B2 e Colimator

" 8 x Wire Scanners e — T ol e o . |
e — e o] BN S S e

Low Energy Beam Distribution Lines: @—” :[ v | erssorsen pxersresomomosn |

= 26 x Faraday Cups fi: ; iy r— ——— T~ |

= 18 x Wire Scanners nex R— N = -
b T wolforr g &l e

REX: o D =] i

= 8 x Faraday Cups o R ] s

* 1 x Collimators/Beam Attenuators Wheel | oo mmsesssmmmmm o - ==

HIE-ISOLDE: — ERE

= 18 x Faraday Cups | o w‘

* 18 x Scanning Slits T e

= 18 x Beam Collimators | - -

= 6 x Stripping Foils |

= 5 x Silicon Detectors e | T

JA Rodriguez
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ISOLDE

ISOLDE BEAM DIAGNOSTICS Main Issues

YGPS BSGA950'-1SOUSERTALL - (INCA)
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showret. || tiderer. || SaveAsRer. |[Sensitiviey  [v]s_200pa Bt | [ showret. |[ mderer |[ saveAsRer. | Sensiciviy  [ws1na Exit SMGS brOken Wires:
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204 1 -
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GRID IN | ouT | | elogBook Autoscale GRID IN out | | elogBook Autoscale
I e = | lo Exception to display...
Scanner Wires Low-Level Control Software:
= Calibration of the the wire scanners after separator dipoles no longer
possible. Functionality lost due to changes in the low-level control B ;inzom.«n:» voor
SOftwa re | Analysis Zoom | |IN [ stan Stop | Slow [w|NO elogBook Autoscale
. . No Exception to display...
= Only four scanners can be running at the same time \

Non-Reliable Collimators/Beam Attenuators Wheel:
» Often skips positions ending up in the wrong attenuation or
collimatorsize (high risk of damaging the Silicon detectors)

ssssssssss

JA Rodriguez
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AD

High Level Interfaces

ADE Fixdisplay 14-Jun-2017 09:34:37
Mode
PBARPROD
No. of Inj. . . .
VISTAR is the most important control interface.
- 1
‘ Shottky real time FFT is still one of the best
AD Cycle Length . .
way to visually detects losses / issues.
Repetition Rate
12s
CPS BCT9012 BCT9053 3.5 GeV/c 2 GeV/c 300 MeV/c

14475 1414.4847 | 1318.9225 [3.24E7 100%|2.95E7 91 %268 E7 82 % BCT 7049 is The Reference for physics butitis
100 MeV/c R[100 MeV/cE| BCT7049 Spare | Spare | Spare

2.74 E7 84%|2.67 E7 82 %| 3.20 E7 98 % NYI NYI NYI very optimistic.
PBAR forsELEW

Comments (13-Jun-2017 09:48:40) _ : :
AD working hours: Lars 163214 / 76688 New d'?g leg DI'BCT',6052 Is still r.10t .
operational due to signal reflectionsissues. It
could be really helpful to tune the AD-target
Settings.

ELENA : Only working hours 76688
No Message

Clobal-State

AD STATUS

EEETEEEEE Inspector Interface checks hundreds of
conditions and allows to control / restart most

E-COOLER Stochastic Cooling [JIMAINS Power Supplies Injection Line Ejection Line Beam Stoppers Cavities C10 & C02

- - --MM of the HW without the need of external tool.
Beam Quality XGEMs & MTVs External Conditions Scrappers Parked

Bl Day 29/06/2017
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AD

BTVs

* 11 BTV systems on injection and ejection lines at
the AD — as well as one at septum inring.

* Working well.

BCCCA

—— Ibeam [uA] 01 —— Ibeam [uA]
ﬁ After a few hick-ups this system is now working
[} 20&00 40600 60600 80609 100000 ° 0 20000 40000 60000 80000 100000 - . - . .
: — st | [ . | et very well and is being used as a major diagnostic
| :::::::ﬂi:l{ou;]"u ] frv-, . ::;;:tatl[x;llilglﬂ . tool.
3 . S —> VERY USEFULL, especially during startup !
P \1 ] Will be incorporated into VISTAR later!
[} 20000 40000 W 80000 100000 ¢ 7 20600 400‘00 60600 80600 100‘000

Losses / optimized

Shottky

- Working pretty well, but also in need
of acquisition system upgrade.
- Planned for LS2

B. Lefort
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AD

Scraper
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= - = = = s o (ADR4SGDOAT DD SPOTEETNRERR)
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E 204
2 (o P S ¥ 050 Pt o N o M
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30 % /RS = 813]
E 204
T 20

E G PREPNI == A, POt X .o ‘_,»'\‘ i
£af o e §‘°|I|I|II|
> -20 N | ainlil (19 |

: I

o s 10 15 20 25 o 5' 10 l‘s 2'0 Z'S 3b
Monitor V Monitor No.

Orbit system and YASP finally working - nearly. Still a few bugs to work out!
Measurement is possible on Plateaux & Rampes.

B. Lefort
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AD

NON.MULTIPLEXED_AD 01 ADE | UNDEFINED

Tune

BBQ system not yet
operational.

The AD presently has
no tune measuring
systemas the old
system was
dismantled to make
way for the BBQ
system!

GEMs

Perfect!

Electron Cooler

* AD cooler is working extremely well!
* Position pickups not working

- AD electron cooler is getting old! 40 years next
year!

 Has worked at ICE, LEAR and AD.

» Electron cooler technology has developed over the
last 40 years!

« Spare parts are getting difficult to get.

. Ara/ major problem can easily lead to down times of
years!

« Planning a new electron cooler for the AD as part of
the AD consolidation program.

* Hope to go to tendering at the end of this year.

B. Lefort
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Summary

instrumentation for the PS complex is complex! (beam parameters spanning
orders of magnitude; large number of users; historic accelerator hardware)
LIU upgrade sets new challenges

we appreciate the constant support from Bl to cope with this
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